Author's response to reviews

Title: A prospective study to assess the value of MMP-9 in improving the appropriateness of urgent referrals for colorectal cancer

Authors:

Angela V Ryan (a.v.ryan@bham.ac.uk)
Sue Wilson (s.wilson@bham.ac.uk)
Michael JO Wakelam (m.j.o.wakelam@bham.ac.uk)
Sally A Warmington (s.a.warmington@bham.ac.uk)
Janet A Dunn (j.a.dunn@warwick.ac.uk)
Richard FD Hobbs (f.d.r.hobbs@bham.ac.uk)
Ashley Martin (a.martin@bham.ac.uk)
Tariq Ismail (tariq.ismail@uhb.nhs.uk)

Version: 2 Date: 15 September 2006

Author's response to reviews:

As requested, results of the pilot study have been removed from the manuscript.

I would be grateful if this protocol could now be considered for publication.